\begin{thebibliography}{10}

\bibitem{marshall}
H.~Jones and J.~Marshall, ``{Convection with rotation in a neutral ocean: a
  study of open deep convection},'' {\em {Journal of Physical Oceanography}},
  vol.~23, no.~6, pp.~1009--1039, 1993.

\bibitem{GCM}
Arvind, K.-C. Cho, C.~Hill, R.~P. Johnson, J.~Marshall, and A.~Shaw, ``{A
  Comparison of Implicitly Parallel Multithreaded and Data Parallel
  Implementations of an Ocean Model based on the Navier-Stokes Equations},''
  Tech. Rep. CSG Memo 364, Laboratory for Computer Science, MIT, 1991.
\newblock {\small http://www-csg.lcs.mit.edu/GCM/}.

\bibitem{id}
R.~S. Nikhil, ``{Id Language Reference Manual, Version 90.1},'' Tech. Rep. CSG
  Memo 284-2, Laboratory for Computer Science, MIT, July 1991.

\bibitem{monsoon}
G.~M. Papadopoulos, {\em {Implementation of a General-Purpose Dataflow
  Multiprocessor}}.
\newblock Research Monograph in Parallel and Distributed Computing, MIT Press,
  1992.

\bibitem{frame-manager}
D.~Chiou, ``{Activation Frame Memory Management for the Monsoon Processor},''
  Master's thesis, MIT EECS, September 1992.
\newblock {\small http://www-csg.lcs.mit.edu/$\sim$derek/MS-Thesis.ps}.

\bibitem{Culler90}
D.~E. Culler, {\em {Managing Parallelism and Resources in Scientific Dataflow
  Programs}}.
\newblock PhD thesis, MIT EECS, 545 Technology Square, Cambridge, MA 02139,
  1990.
\newblock LCS TR-446.

\bibitem{Wolfe95}
M.~Wolfe, {\em High Performance Compiler for Parallel Computing}.
\newblock Addison Wesley, 1995.

\bibitem{boon-loops}
B.~S. Ang, ``Efficient implementation of sequential loops in dataflow
  computation,'' in {\em Proceedings of the Functional Programming Computer
  Architecture}, June 1993.
\newblock {\small ftp://csg-ftp.lcs.mit.edu/pub/papers/misc/ seq-loop.ps.Z}.

\bibitem{templates}
R.~Barrett, M.~Berry, T.~F. Chan, J.~Demmel, J.~Donato, J.~Dongarra,
  V.~Eijkhout, R.~Pozo, C.~Romine, and H.~Van~der Vorst, {\em Templates for the
  Solution of Linear Systems: Building Blocks for Iterative Methods}.
\newblock Philadelphia, PA: SIAM, 1994.
\newblock {\small http://www.netlib.org/linalg/html\_templates/
  Templates.html}.

\bibitem{CMF}
Thinking Machine Corporation, {\em CM Fortran Language Reference Manual, V2.1},
  January 1994.

\bibitem{CM-5}
W.~D. Hillis and L.~W. Tucker, ``The {CM-5} connection machine: {A} scalable
  supercomputer,'' {\em Communications of the ACM}, vol.~36, pp.~31--40, Nov.
  1993.

\bibitem{shaw92b}
A.~Shaw, Y.~Kodama, M.~Sato, S.~Sakai, and Y.~Yamaguchi, ``{Performance of
  Data-Parallel Primitives on the EM-4 Dataflow Parallel Supercomputer},'' in
  {\em Proceedings of Frontiers '92}, (McLean, VA), pp.~302--309, October 1992.

\bibitem{EM4a}
S.~Sakai, Y.~Yamaguchi, K.~Hiraki, Y.~Kodama, and T.~Yuba, ``{An Architecture
  of a Dataflow Single Chip Processor},'' in {\em Proceedings of ISCA},
  pp.~46--53, 1989.

\bibitem{Grafe88a}
V.~Grafe, G.~Davidson, J.~Hoch, and V.~Holmes, ``{The Epsilon Dataflow
  Processor},'' in {\em Proceedings of the 16th. Annual International Symposium
  on Computer Architecture, Jerusalem, Israel}, pp.~36--45, May 29-31 1989.

\bibitem{Mohr91}
E.~Mohr, D.~Kranz, and R.~Halstead~Jr., ``{Lazy task creation a technique for
  increasing the granularity of parallel programs},'' {\em IEEE Transactions on
  Parallel and Distributed Computing}, vol.~2, pp.~264--80, July 1991.
\newblock {\small ftp://cag.lcs.mit.edu/pub/papers/futures.ps.Z}.

\bibitem{Poly87}
C.~D. Polychronopoulos and D.~J. Kuck, ``{Guided self-scheduling: a practical
  scheduling scheme for parallel supercomputers},'' {\em {IEEE Transactions on
  Computers}}, vol.~C-36, pp.~1425--39, December 1987.

\bibitem{Arvind88b}
Arvind, D.~E. Culler, and K.~Ekanadham, ``The price of asynchronous
  parallelism: an analysis of dataflow architectures,'' in {\em Proceedings of
  CONPAR 88}, pp.~541--555, Sept. 1988.

\bibitem{id-performance}
J.~Hicks, D.~Chiou, B.~S. Ang, and Arvind, ``{Performance Studies of the
  Monsoon Dataflow Processor},'' {\em Journal of Parallel and Distributed
  Computing}, vol.~18, pp.~273--300, July 1993.
\newblock {\small
  http://www-csg.lcs.mit.edu/$\sim$derek/Monsoon-Performance.ps}.

\bibitem{Yeung93}
D.~Yeung and A.~Agarwal, ``{Experience with Fine-Grain Synchronization in MIMD
  Machines for Preconditioned Conjugate Gradient},'' in {\em {Proceedings of
  the Fourth Symposium on Principles and Practices of Parallel Programming}},
  pp.~{187--197}, {May} 1993.
\newblock {\small ftp://cag.lcs.mit.edu/pub/papers/fine-grain.ps.Z}.

\bibitem{Sur94}
S.~Sur and W.~Bohm, ``{Functional, I-structure, and M-structure implementations
  of NAS benchmark FT},'' in {\em Parallel Architectures and Compilation
  Techniques}, pp.~47--56, August 1994.
\newblock {\small http://www.colostate.edu/ $\sim$dataflow/papers/pact94b.ps}.

\end{thebibliography}

\section{Conclusion}
\label{section:conclusion}

Tuning for dataflow execution (either by the programmer or compiler)
requires techniques which are different from tuning on conventional
architectures.  In this paper, we have characterized some of the overheads
of dataflow execution on Monsoon, focusing on loops and using an example of
a preconditioned conjugate gradient algorithm (PCG).  Our performance
measured by {\it cycles executed per required floating point operation} is
comparable to a production version of the algorithm running on the CM-5.

Much of the overhead comes from the dataflow switch and fork instructions,
as well as bubbles which result from fine-grained synchronization on
Monsoon.  Part of this overhead could be eliminated by architectural
changes -- for instance, the EM-4 \cite{EM4a} architecture uses essentially
the same dataflow-style synchronization mechanism as Monsoon, but does not
incur a bubble because the synchronization and execution units are
decoupled.  In addition, due to some hardware limitations, some idle cycles
on Monsoon are the result of network access contention, rather than being
fundamentally associated with dataflow architectures.  The fork overhead is
eliminated in the Epsilon dataflow architecture \cite{Grafe88a} with a {\it
copy} instruction which can send values to multiple instructions --
however, the {\it copy} instruction reduces instruction count, but does not
reduce cycle count.

Some overhead is the result of the $k$-bounded loop scheme, which has
proven to be difficult to use for the programmer as well as incurring
significant execution overhead.  Setting $k$ for the $k$-bounded loop is a
non-trivial task, and does not lend itself easily to compiler analysis.
Whereas $k$-bounding assumes an eager model of parallelism exposure,
run-time, demand-based approaches such as lazy task creation \cite{Mohr91}
or guided self-scheduling \cite{Poly87} may prove more effective, as well
as easier to program -- however, in general, these techniques are difficult
to implement on Monsoon because it is almost impossible to determine when a
Monsoon processor is idle because of the deeply-pipelined,
heavily-interleaved model of execution.  Again, this is not an inherent
feature of dataflow architectures, but an oversight in the hardware design
of Monsoon.

Language definition and compiler optimization also affect performance.
Id's policy of array extents being defined only at run-time inhibits some
standard loop optimizations, and may be changed by either changing the
language or adding pragmas.  Some compiler loop optimizations for
conventional architectures may not be effective for Monsoon because the
addition of induction variables is very expensive for the dataflow loop
schemas.  However, some other loop optimizations may be more effective for
Monsoon than conventional architectures because of the high overhead of
dataflow loop execution.

The programmer plays a large part in attaining good performance.  Loop
schemas and $k$-bounds should be chosen to expose just enough parallelism
to keep the machine busy.  Changes to the algorithm in order to take
advantage of complete loop unfolding, such as the hybrid version of the
PCG, can also improve performance in non-intuitive ways.

In related work, Arvind, et.al.~\cite{Arvind88b} gives a more detailed
analysis of the overheads of dataflow execution.  Hicks,
et.al.~\cite{id-performance} describes performance of several smaller
benchmarks on Monsoon.  Yeung and Agarwal \cite{Yeung93} discuss the
importance of language support for the expression of fine-grained
parallelism using preconditioned conjugate gradient (primarily the
preconditioning step, which is significantly different from ours) on
Alewife.  Sur and Bohm \cite{Sur94} have performed many performance
analyses of scientific codes in Id on Monsoon, primarily focusing on
language features and their impact on performance.

Although it is unlikely that dataflow architectures (as they exist today)
will be viable commercially in the near future, multithreading techniques
and parallelism are playing an increasingly important role in the design of
commercial systems.  The synchronization and control overheads we encounter
in dataflow computing are inherent to fine-grained parallel multithreaded
execution, and these overheads change the relative importance of program
optimizations, motivating increases in sizes of basic blocks through
inlining, loop unrolling, and program structuring.  The higher overhead of
control instructions also motivates decreases in the sizes of basic block
interfaces, even when the size of the basic block may be decreased by
adding a basic block interface.
